Original Study

Venetoclax and Cobimetinib in Relapsed/Refractory AML: A Phase 1b Trial

Marina Y. Konopleva,^{1,#} Monique Dail,^{2,#} Naval G. Daver,^{1,#} Jacqueline S. Garcia,³
Brian A. Jonas,⁴ Karen W.L. Yee,⁵ Kevin R. Kelly,⁶ Norbert Vey,⁷ Sarit Assouline,⁸
Gail J. Roboz,⁹ Stefania Paolini,¹⁰ Daniel A. Pollyea,¹¹ Agostino Tafuri,¹²
Joseph M. Brandwein,¹³ Arnaud Pigneux,¹⁴ Bayard L. Powell,¹⁵ Pierre Fenaux,¹⁶
Rebecca L. Olin,¹⁷ Giuseppe Visani,¹⁸ Giovanni Martinelli,¹⁹ Maika Onishi,²
Jue Wang,² Weize Huang,² Diana R. Dunshee,² Habib Hamidi,² Marion G. Ott,²⁰
Wan-Jen Hong,²¹ Michael Andreeff¹

Abstract

Treatment options for patients who have acute myeloid leukemia (AML) which has come back (relapsed) or stopped responding to treatment (refractory) are limited. In this study, 30 patients with relapsed/refractory AML received 2 drugs (venetoclax and cobimetinib). Venetoclax-cobimetinib had limited responses with unwanted side effects, compared with venetoclax alone. However, these findings will help future trials of similar drug combinations.

Background: Therapies for relapsed/refractory acute myeloid leukemia remain limited and outcomes poor, especially amongst patients who are ineligible for cytotoxic chemotherapy or targeted therapies. **Patients and Methods:** This phase 1b trial evaluated venetoclax, a B-cell lymphoma-2 (BCL-2) inhibitor, plus cobimetinib, a MEK1/2 inhibitor, in patients with relapsed/refractory acute myeloid leukemia, ineligible for cytotoxic chemotherapy. Two-dimensional dose-escalation was performed for venetoclax dosed daily, and for cobimetinib dosed on days 1-21 of each 28-day cycle. **Results:** Thirty patients (median [range] age: 71.5 years [60-84]) received venetoclax-cobimetinib. The most common adverse events (AEs; in \geq 40.0% of patients) were diarrhea (80.0%), nausea (60.0%), vomiting (40.0%), febrile neutropenia (40.0%), and fatigue (40.0%). Overall, 66.7% and 23.3% of patients experienced AEs leading to dose modification/interruption or treatment withdrawal, respectively. The composite complete remission (CRc) rate (complete remission [CR] + CR with incomplete blood count recovery + CR with incomplete platelet recovery) was 15.6%; antileukemic

Abbreviations: AE, adverse event; AHD, antecedent hematologic disorder; AML, acute myeloid leukemia; BCL-2, B-cell lymphoma-2; BCL-xL, B-cell lymphoma-extra large; CI, confidence interval; C_{max}, maximum plasma concentration at steady state; CR, complete remission; CRc, composite CR; CRi, CR with incomplete blood count recovery; CRp, CR with incomplete platelet count recovery; HSCT, hematopoietic stem cell transplant; MAPK, mitogen-activated protein kinase; MEK1/2, MAPK kinases 1 and 2; MLFS, morphologic leukemia-free state; MR, minor responses; MTD, maximum tolerated dose; OS, overall survival; PD, pharmacodynamic; pERK, phosphorylated ERK; PK, pharmacokinetic; PR, partial response; pS6, phosphorylated S6; R/R, relapsed or refractory; RP2D, recommended phase 2 dose; SAE, serious adverse event; sAML, secondary AML; SD, stable disease; TLS, tumor lysis syndrome; VAF, variant allele frequency; ven-cobi, venetoclax and cobimetinib; ven-idasa, venetoclax in combination with idasanutlin.

- ¹University of Texas, MD Anderson Cancer Center, Houston, TX
- ²Genentech, Inc., South San Francisco, CA
- ³Dana-Farber Cancer Institute, Boston, MA
- ⁴University of California Davis Comprehensive Cancer Center, Sacramento, CA
- ⁵Princess Margaret Cancer Centre, Toronto, ON, Canada
- ⁶University of Southern California, Los Angeles, CA
- ⁷Hematologie Clinique, Institut Paoli Calmettes, Marseille, France
- ⁸Jewish General Hospital, Montreal, QC, Canada
- ⁹Weill-Cornell Medical College, New York Presbyterian, New York, NY
- ¹⁰IRCCS Azienda Ospedaliero-Universitaria di Bologna, Istituto di Ematologia "Seràgnoli", Bologna, Italy
- ¹¹University of Colorado, Division of Hematology, Aurora, CO

- ¹²Department of Clinical and Molecular Medicine, University Hospital Sant'Andrea-Sapienza, Rome, Italy
- ¹³Division of Hematology, University of Alberta, Edmonton, AB, Canada
- ¹⁴Bordeaux Haut-Leveque University Hospital, Pessac, France

- ¹⁶Hôpital Saint-Louis, Université Paris Diderot, Paris, France
- ¹⁷University of California San Francisco, San Francisco, CA
- ¹⁸Hematology, Ospedale San Salvatore, Pesaro, Italy
- ¹⁹IRCCS Istituto Romagnolo per lo Studio dei Tumori (IRST) "Dino Amadori",
- Meldola, Italy
- ²⁰F. Hoffmann-La Roche Ltd, Basel, Switzerland
- ²¹Prelude Therapeutics, Wilmington, DE

2152-2650/\$ - see front matter © 2024 The Authors. Published by Elsevier Inc.This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/) https://doi.org/10.1016/j.clml.2024.01.007

¹⁵Atrium Health Wake Forest Baptist Comprehensive Cancer Center, Winston-Salem, NC

Submitted: Nov 14, 2023; Revised: Jan 15, 2024; Accepted: Jan 15, 2024; Epub: 18 January 2024

Address for correspondence: Marina Y. Konopleva, Department of Leukemia, Division of Cancer Medicine, University of Texas MD Anderson Cancer Center, 1400 Holcombe Boulevard, Houston, TX, 77030. E-mail contact: mkonole@mdanderson.org

[#] Authors contributed equally.

response rate (CRc + morphologic leukemia-free state/partial remission) was 18.8%. For the recommended phase 2 dose (venetoclax: 600 mg; cobimetinib: 40 mg), CRc and antileukemic response rates were both 12.5%. Failure to achieve an antileukemic response was associated with elevated baseline phosphorylated ERK and MCL-1 levels, but not BCL-xL. Baseline mutations in \geq 1 signaling gene or *TP53* were noted in nonresponders and emerged on treatment. Pharmacodynamic biomarkers revealed inconsistent, transient inhibition of the mitogen-activated protein kinase (MAPK) pathway. **Conclusion:** Venetoclax-cobimetinib showed limited preliminary efficacy similar to single-agent venetoclax, but with added toxicity. Our findings will inform future trials of BCL-2/MAPK pathway inhibitor combinations.

 Clinical Lymphoma, Myeloma and Leukemia, Vol. 24, No. 6, 364–374 © 2024 The Authors. Published by Elsevier Inc. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/)
 Keywords: B-cell lymphoma-2, Biomarker, Mitogen-activated protein kinase pathway, Targeted therapy

Introduction

Acute myeloid leukemia (AML) is an aggressive malignancy, and largely a disease of older patients, with a median age at diagnosis of 68 years.¹ Therapies for patients with relapsed or refractory (R/R) AML remain limited and outcomes poor, especially for patients who are ineligible for cytotoxic chemotherapy or targeted therapies.²⁻⁶ Therapies that target survival pathways known to be dysregulated or aberrant in AML may improve outcomes, particularly if they are less toxic.

Venetoclax, an oral B-cell lymphoma-2 (BCL-2) inhibitor,⁷ has shown limited activity as a monotherapy in R/R AML, with a complete remission (CR)/CR with incomplete blood count recovery (CRi) rate of 19%.⁸ Cobimetinib, a small molecule inhibitor of mitogen-activated protein kinase (MAPK) kinases 1 and 2 (MEK1/2),⁹ is approved for use in melanoma^{10,11} but has not been evaluated in AML. Other MEK1/2 inhibitors, such as trametinib, selumetinib, and binimetinib, however, have shown limited single agent activity in patients with *RAS*-mutated R/R AML and myelodysplastic syndromes, with response rates (with variable definitions) of 20%, 17%, and 8%, respectively.¹²⁻¹⁴

Concomitant BCL-2 and MAPK/MEK blockade has demonstrated synergistic induction of apoptosis in *in vitro* AML models, including those resistant to single agents, and reduced leukemia burden in *in vivo* xenograft models.¹⁵⁻¹⁸ For instance, the combination of venetoclax and cobimetinib (ven-cobi) significantly enhanced cell death, and suppressed cell growth and the clonogenic potential of myeloid progenitors, compared with venetoclax or cobimetinib alone.¹⁵ This synergy has also been associated with MCL-1 downregulation, upregulation of pro-apoptotic BIM, and suppression of phosphorylated S6 (pS6).¹⁵⁻¹⁸

Overexpression of MCL-1 has been identified as a major acquired mechanism of venetoclax resistance, as well as a contributor to AML progression.^{19,20} Thus, a venetoclax combination targeting MAPK, and indirectly MCL-1, may be a potential option to improve AML therapy. We therefore initiated a phase 1 study to assess the safety, tolerability, and preliminary efficacy of ven-cobi in older patients with R/R AML, who were ineligible for cytotoxic chemotherapy.

Methods

Study Design

This open-label, multicenter phase 1b trial evaluated ven-cobi and venetoclax in combination with idasanutlin (ven-idasa; ClinicalTrials.gov identifier: NCT02670044). Here, we present results for the ven-cobi arm. Details of the ven-idasa arm have been previously published.²¹ Primary objectives included: safety profile evaluation and determination of the maximum tolerated dose (MTD) and recommended phase 2 dose (RP2D). Secondary objectives were to evaluate preliminary efficacy and pharmacokinetics (PK). Exploratory objectives included assessing biomarkers related to drug targets, cobimetinib exposure, and disease biology.

Enrollment was nonrandomized, with multiple cohorts open simultaneously, and patients were only treated with 1 combination arm (ven-cobi or ven-idasa). There were 4 ven-cobi cohorts in total: venetoclax 400 mg + cobimetinib 40 mg (Cohort 1); venetoclax 600 mg + cobimetinib 40 mg (Cohort 2); venetoclax 800 mg + cobimetinib 40 mg (Cohort 3); and venetoclax 400 mg + cobimetinib 60 mg (Cohort 4). Venetoclax initiation included a 3- to 5-day ramp-up to 400, 600, or 800 mg daily on days 1-28, with cobimetinib (40 or 60 mg daily) on days 1-21 of each 28-day cycle; dose-escalation followed a 2-dimensional, 3+3+3design (Supplemental Figure S1). Starting doses were established based on the clinical experience with each drug. The starting dose of venetoclax was 400 mg daily, half the venetoclax monotherapy dose tested in a phase 2 study of R/R or previously untreated AML (M14-212), in which doses of up to 1200 mg daily were well tolerated.8 The starting dose of cobimetinib was 40 mg daily, 1 dose level below the dose of cobimetinib used in the phase 3 study of cobivemurafenib in metastatic melanoma (60 mg once daily; coBRIM study, GO28141).²² Patients were hospitalized during ramp-up in cycle 1 and received tumor lysis syndrome (TLS) prophylaxis. Safety monitoring criteria and stopping rules for toxicity were protocoldefined.

The study protocol was approved by the institutional review board or ethics committees at participating institutions in accordance with the International Conference on Harmonization guidelines, including good clinical practice and the ethical principles originating from the Declaration of Helsinki. Informed consent was obtained from all patients. Authors had access to and reviewed the clinical trial results.

Patient Population

Eligible patients had R/R AML by World Health Organization criteria²³ or newly diagnosed secondary AML (sAML) after prior treatment for an antecedent hematologic disorder (AHD).

Patients were deemed ineligible for cytotoxic chemotherapy based on age (\geq 60 years) and investigator opinion. Key exclusion criteria included the use of strong or moderate cytochrome P450-3A inducers or inhibitors \leq 7 days before study drug administration; prior use of a BCL-2 inhibitor; or prior exposure to experimental treatment targeting the Raf, MEK, or MAPK pathways.

Assessments

Adverse events (AEs) were reported by the treating physician throughout the study and for \geq 30 days after the last dose or initiation of another anticancer therapy. MTD was the highest dose at which less than one-third of \geq 6 patients experienced a dose-limiting toxicity.

Response was assessed by routine laboratory tests and bone marrow examinations and evaluated per the International Working Group 2003 AML response criteria.²⁴ Composite CR (CRc) was defined as CR + CRi + CR with incomplete platelet count recovery (CRp), and antileukemic response as CRc + morphologic leukemia-free state (MLFS)/partial response (PR). CRp and CRi were considered mutually exclusive.

Pharmacokinetic Assessments

Plasma concentration-versus-time data for venetoclax and cobimetinib were analyzed using noncompartmental analysis. Summary statistics of PK parameters, such as maximum plasma concentration at steady state (C_{max}) and trough/minimum plasma concentration at steady state, of venetoclax and cobimetinib were computed.

Biomarker Assessments

Mutation analysis was performed on bone marrow-derived mononuclear cells at baseline, at the end of the study and, in a few cases, during treatment, using the FoundationOne Heme Panel (465 gene mutation panel, Roche Foundation Medicine, Grenzach-Wyhlen, Germany) as previously published.²⁵

Intracellular protein expression of BCL-2, MCL-1, BCL-xL, phosphorylated ERK (pERK), and pS6 was evaluated on blast cells, myeloid/monocytes, and lymphocytes using surface backbone lineage markers to identify cell populations (CD45, CD34, CD117, HLA-DR). Baseline expression of BCL-2, BCL-xL, and MCL-1 in blast cells was evaluated for association with response; since high MCL-1 and/or high BCL-xL are common resistance mechanisms for venetoclax,²⁶ ratios of BCL-2:BCL-xL and BCL-2:MCL-1 were evaluated.

Additional details of further assessments performed are provided in the Supplementary Methods.

Statistical Methods

Safety and efficacy were summarized by descriptive statistics. The efficacy population was the intent-to-treat population; the safety-evaluable population included all patients who received one or more dose(s) of study drug. Time-to-event analyses were conducted using the Kaplan–Meier method, and the Fisher's exact test compared response rates between treatment groups.

Results

A total of 32 patients were enrolled in the ven-cobi arm across 17 centers (United States, Italy, Canada, and France) between March 2016 and July 2020 (Supplemental Figure S2). Data cutoff was December 10, 2020, and median (range) duration of follow-up was 2.7 months (0.0-14.8). Overall, 30 patients received one or more dose(s) of study drug, and the median (range) number of cycles received was 2 (1-8).

Patient Characteristics

At baseline, median (range) patient age was 71.5 years (60-84), 63.3% of patients had refractory disease, and 36.7% had relapsed disease; no patients had newly diagnosed AML transformed from a previously treated AHD. Half (50.0%) of patients had sAML. Patients had received a median (range) of 2 prior lines of therapy (1-10), which included azacitidine (36.7% of patients) and decitabine (16.7% of patients) (Supplemental Table S1); 20.0% of patients had received a prior hematopoietic stem cell transplant (HSCT); and 33.3% of patients were adverse risk by European LeukemiaNet 2010 criteria. Baseline cytogenetics included 23.3% of patients with complex karyotype, and mutation analysis identified *NRAS* and/or *KRAS* mutations in 23.3% of patients and *TP53* mutations in 13.3% of patients (Table 1).

Safety

The most common (occurring in \geq 40.0% of patients) allgrade treatment-emergent AEs were diarrhea (80.0%), nausea (60.0%), vomiting (40.0%), febrile neutropenia (40.0%), and fatigue (40.0%) (Supplemental Table S2A). The most common (occurring in \geq 20.0% of patients) grade 3-4 AEs included febrile neutropenia (40.0%), diarrhea (36.7%), and pneumonia (30.0%). Diarrhea (grade 3 only; no grade 4) was most commonly experienced at higher doses: 25.0% of patients in Cohort 1; 28.6% of patients in Cohort 2; 33.3% of patients in Cohort 3; and 57.1% of patients in Cohort 4 (Supplemental Table S2B). Nearly all cases (49/52) resolved completely. Due to the frequent occurrence of diarrhea, mandatory antidiarrheal prophylaxis was implemented as a protocol amendment; however, no patients were treated with ven-cobi following the amendment due to the Sponsor's decision to not pursue further investigation of the combination given the overall benefit-risk profile. Although the number of patients in each cohort was small, there were also higher rates of febrile neutropenia in Cohorts 3 (66.7%) and 4 (42.9%) compared with Cohorts 1 (0%) and 2 (14.3%). Serious AEs (SAEs) were reported in 80.0% of patients. The most commonly reported SAEs (occurring in \geq 20.0% of patients) were febrile neutropenia (30.0%), pneumonia (30.0%), and sepsis (20.0%) (Supplemental Table S3). Fatal AEs were reported in 6 (20.0%) patients and consisted of: lung infection (n = 2), sepsis (n = 2), respiratory failure (n = 1), and lung disorder (n = 1).

Overall, 66.7% of patients experienced an AE that resulted in a treatment modification or interruption, most commonly (\geq 10.0% of patients) due to diarrhea (26.7%), febrile neutropenia (13.3%), and thrombocytopenia (13.3%) (Supplemental Table S4 and Supplemental Figure S3A-B). Treatment withdrawal due to

of the Safety Population.					
	n = 30				
Median age, y (range)	71.5 (60-84)				
Male sex, n (%)	16 (53.3)				
ECOG PS, n (%) 0 1 2	5 (16.7) 20 (66.7) 5 (16.7)				
Disease status, n (%) Refractory Relapsed Newly diagnosed, transformed from AHD (previously treated)	19 (63.3) 11 (36.7) 0 (0.0)				
AML type, n (%) <i>De novo</i> Secondary	15 (50.0) 15 (50.0)				
ELN 2010 classification, n (%) Favorable Intermediate-I Intermediate-II Adverse	1 (3.3) 9 (30.0) 10 (33.3) 10 (33.3)				
Median prior therapies, n (range)	2 (1-10)				
Prior HSCT, n (%)	6 (20.0)				
WBC at baseline, 10 ⁹ /L, median (range)	2.1 (0.5-23.2)				
Aspirate BM blast %, median (aspirate)	30 (7.0-96.0)				
Mutations and cytogenetics, n (%)					
NRAS/KRAS mutation status ^a NRAS and/or KRAS mutation detected NRAS mutation detected KRAS mutation detected NRAS and KRAS mutation undetected Not evaluable	7 (23.3) 6 (20.0) 3 (10.0) 21 (70.0) 2 (6.7)				
<i>TP53</i> mutation status ^b Mutation detected Mutation undetected Not evaluable	4 (13.3) 24 (80.0) 2 (6.7)				
t(9;11)(p22;q23); <i>MLLT3-MLL</i>	1 (3.3)				
inv(3)(q21q26.2) or t(3;3)(q21;q26.2); <i>RPN1-EVI1</i>	1 (3.3)				
-5 or del(5q)	1 (3.3)				
-7	4 (13.3)				
Cytogenetic abnormalities not classified as favorable or adverse	is not classified as 10 (33.3)				
Complex karyotype	7 (23.3)				

Detient Demographics and Deceline Characteristic

Abbreviations: AHD = antecedent hematologic disorder; AML = acute myeloid leukemia; BM = bone marrow; ECOG PS = Eastern Cooperative Oncology Group performance status; ELN = European LeukemiaNet; HSCT = hematopoietic stem cell transplantation; WBC = white blood cells.

^a Using FM Heme cutoff of 5%.

^b Using FM Heme cutoff of 1%.

an AE or SAE occurred in 23.3% and 20.0% of patients, respectively; specific treatment-emergent AEs leading to any treatment withdrawal are available in Supplemental Table S5. No AEs, including diarrhea, resulted in treatment discontinuation in more than 1 patient (Supplemental Table S6).

Marina Y. Konopleva et al

Laboratory TLS and clinical TLS were reported in 1 patient each; the clinical TLS (grade 3) event was associated with an increase in creatinine in the setting of pneumonia and was managed with standard measures. No TLS events resulted in treatment discontinuation. Overall, 30- and 60-day Kaplan–Meier estimated mortality rates were 6.7% and 34.3%, respectively. Causes of 60-day mortality included: progressive disease (6/30 patients); respiratory failure (1/30; preceded by pneumonia in the resistant disease setting, assessed as unrelated to study treatment); sepsis (2/30; gramnegative sepsis, assessed as unrelated to study treatment; sepsis, assessed as related to study treatment); and lung disorder (1/30; lung infection in the resistant disease setting, assessed as related to study treatment).

Dose-limiting toxicities reported were diarrhea (2 patients [6.7%]: 1 each in Cohorts 2 and 4), and ejection fraction decreased (2 patients [6.7%]: 1 each in Cohorts 3 [assessed as related to both venetoclax and cobimetinib] and 4 [assessed as unrelated to venetoclax and cobimetinib]). Cohort 2 (venetoclax 600 mg + cobimetinib 40 mg) was determined to be the MTD and RP2D. Although the MTD was not exceeded in Cohort 3, Cohort 2 was established as the RP2D due to the overall safety profile.

Efficacy

CRc and antileukemic response rates across all dose cohorts were 15.6% (5/32) and 18.8% (6/32), respectively. At the RP2D, the CRc and antileukemic response rates were both 12.5% (1/8). The median (range) time to best CRc response was 1.8 months (1.4-3.0), and the median (range) duration of response was 4.2 months (1.0-10.4) (Table 2). Blast count reduction was seen across all ven-cobi dose cohorts (Figure 1). Responses were seen in patients aged \geq 75 years, patients with sAML, and patients with AML who had received prior treatment with a hypomethylating agent (Supplemental Figure S4).

The median (95% confidence interval [CI] overall survival [OS] was 3.6 months [2.0-8.6] among all patients, and 7.5 months (4.2 months-not evaluable) among patients with a CRc response (Supplemental Figure S5). No evaluable patients with CRc achieved minimal residual disease negativity ($<10^{-3}$ threshold). No patients underwent a subsequent HSCT.

While minimal residual disease was not achieved, molecular responses (reductions in baseline mutation variant allele frequency [VAF]) in responders were noted when evaluating on-treatment changes to baseline mutation frequencies: VAF reductions were observed in patients who achieved CR (mean change: -11.3%) compared with patients with stable disease (SD; mean change: +2.8%) or refractory disease (mean change: +11.7%) (Supplemental Figure S6A-B). Changes were similar in genes coding for signaling or nonsignaling proteins.

Pharmacokinetics

PK parameters were available for 22 patients across the 4 cohorts. In Cohorts 1 (n = 4), 2 (n = 7), 3 (n = 8), and 4 (n = 3), respectively, the mean C_{max} for venetoclax was 1.34, 1.48, 1.64, and 1.16 µg/mL, and the mean C_{max} for cobimetinib was 0.32, 0.20, 0.39, and 0.64 µg/mL (Supplemental Table S7).

Table 2 Response Outcomes for Ven-Cobi.						
п (%)	Cohort 1: ven 400 mg + cobi 40 mg (n = 4)	Cohort 2: ven 600 mg + cobi 40 mg (n = 8)	Cohort 3: ven 800 mg + cobi 40 mg (n = 12)	Cohort 4: ven 400 mg + cobi 60 mg (n = 8)	Total (N = 32)	
Antileukemic responders (CRc/PR/MLFS)	1 (25.0)	1 (12.5)	3 (25.0)	1 (12.5)	6 (18.8)	
CRc (CR/CRi/CRp)	1 (25.0)	1 (12.5)	2 (16.7)	1 (12.5)	5 (15.6)	
CR	0 (0.0)	1 (12.5)	2 (16.7)	0 (0.0)	3 (9.4)	
CRi	1 (25.0)	0 (0.0)	0 (0.0)	0 (0.0)	1 (3.1)	
CRp	0 (0.0)	0 (0.0)	0 (0.0)	1 (12.5)	1 (3.1)	
PR	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	
MLFS	0 (0.0)	0 (0.0)	1 (8.3)	0 (0.0)	1 (3.1)	
Time to best CRc response (months), median (range)	2.8 (2.8-2.8)	1.4 (1.4-1.4)	2.3 (1.6-3.0)	1.8 (1.8-1.8)	1.8 (1.4-3.0)	
Duration of response (months), median (range)	2.0 (2.0-2.0)	4.4 (4.4-4.4)	2.6 (1.0-4.2)	10.4 (10.4-10.4)	4.2 (1.0-10.4)	
Duration of follow-up (months), median (range)	4.7 (1.6-8.7)	4.8 (0.0-14.2)	3.0 (0.6-14.8)	1.1 (0.0-12.3)	2.7 (0.0-14.8)	

Abbreviations: cobi = cobimetinib; CR = complete remission; CRc = complete remission; CRi = complete remission with incomplete blood count recovery; CRp = complete remission with incomplete platelet recovery; MLFS = morphologic leukemia-free state; PR = partial response; ven = venetoclax.

Figure 1 Best percentage bone marrow blast reduction and individual patient responses. Efficacy data are presented for the intent-to-treat population for whom data were available.



*Best % change from baseline in bone marrow blasts is > 100. Abbreviations: cobi = cobimetinib; CR = complete remission; CRi = complete remission with incomplete blood count recovery; CRp = complete remission with incomplete platelet recovery; MLFS = morphologic leukemia-free state; RD = refractory disease; SD = stable disease; ven = venetoclax.

Pharmacodynamic (PD) Changes

Reductions in MEK downstream proteins pERK and pS6 (>30% reduction [or 0.7-fold change] in expressing cells) were observed in most patients (13/20 for both). pERK reductions were highly variable with 50.0% median reduction at 6-hour and 62.5% median "best" reduction at any timepoint (Supplemental Figure S7A-B). Furthermore, pERK reductions were often observed early, at the 6-hour timepoint (9/14 patients with 6-hour data), but often rebounded by day 5 to baseline values or higher (5/9 patients; Figure 2A and Supplemental Figure S7C); in contrast, pS6 reduc-

tions occurred later, with >30% reductions observed at either days 5 or 15 in most patients (11/21) (Supplemental Figure S7D-E). Notably, there were no dose interruptions associated with the observed rebound of pERK levels.

No significant PK/PD correlations were identified. Neither pERK nor pS6 reductions were associated with dose, cobimetinib exposure, or response (Figure 2A and Supplemental Figures S7D, S8A-D, and S7C and S7E, respectively). Similarly, reductions in MCL-1 and BCL-xL were not associated with dose, exposure, or response (Figure 2B and Supplemental Figures S9A, S8E-H, and S9B, respec-

Marina Y. Konopleva et al





Mutations detected at baseline are indicated by shaded boxes, mutation loss by a diagonal line, and emergent mutations by an asterisk. Abbreviations: BCL-2 = B-cell lymphoma 2; BCL-xL = B-cell lymphoma-extra large; cobi = cobimetinib; cCR = clinical complete response; CR = complete remission; CRi = complete remission with incomplete blood count recovery; CRp = complete remission with incomplete platelet recovery; D = day; FC = fold change; h = hour; MCL-1 = myeloid leukemia 1; MLFS = morphologic leukemia-free state; ND = no data; pERK = phosphorylated ERK; PR = partial response; RD = refractory disease; SD = stable disease.

tively). Further evaluation of longitudinal MCL-1 on treatment revealed inefficient suppression over time (Supplemental Figure S9C).

Correlative Biomarker Analysis

At baseline, pERK was significantly elevated in responders (MLFS+; P = .0076), whereas pS6 was not (Figure 2C and Supplemental Figure S7F). Antileukemic responses were doubled in patients with high (>1.5) BCL-2:BCL-xL and/or BCL-2:MCL-1 (33.3%; 2/6) versus low (\leq 1.5) BCL-2:BCL-xL and BCL-2:MCL-1 (16.7%; 3/18) ratios (Figure 2D). Analysis of each individual BCL-2 family protein in nonresponders revealed high MCL-1 levels, but not BCL-xL (Supplemental Figure S10).

Baseline mutation profiling was available for most patients (28/30); out of the patients that achieved MLFS or better (n = 6), sequencing data were available for only 4 patients. Baseline *TP53* mutations were associated with nonresponse (antileukemic response rate: 0%, 0/4). The antileukemic response rate was 14.3% (1/7) in patients with both baseline *IDH1/2* and *NRAS/KRAS* mutations; 12.5% (2/16) in patients with any signaling mutation (*NRAS, KRAS, FLT3, PTPN11, CBL, JAK2,* or *KIT*); and 0% (0/4) in patients with 2 or more signaling mutations (Figure 2E and Supplemental Table S8).

Emergent mutations, detected on treatment but not at baseline, were also analyzed. At discontinuation, emergent mutations were detected in *TP53* (16.7% of patients, 3/18) and signaling genes (16.7% of patients, 3/18 harboring 4 mutations: *PTPN11* [11.1%, 2/18], *KRAS* [5.6%, 1/18], and *CBL* [5.6%, 1/18]) (Figure 2E and Supplemental Figure S6C). Notably, *CBL* and *PTPN11* mutations both emerged in the same patient, who also harbored a stable *NRAS* mutation and achieved SD. Emergent *TET2* and *IKZF1* mutations were also noted in 1 patient each (5.6% of patients, 1/18) (Figure 2E and Supplemental Figure S6C). Clearance of *NRAS* (and *NPM1)* mutations was observed at the same time as an emergent *PTPN11* mutation (noted above) in a patient that achieved a CR. In addition, *KRAS* mutation clearance occurred in a patient with SD that harbored 3 other stable signaling mutations (Figure 2E and Supplemental Figure S6C).

Gene expression profiling using RNA sequencing was performed on bone marrow aspirates in some patients at baseline (n = 17) and on treatment (n = 26). On-treatment timepoints varied from end of cycle 1 to treatment discontinuation. The "hallmark *KRAS* signaling Up" gene signature²⁷ was significantly increased in patients on or after ven-cobi therapy but not ven-idasa therapy (Supplemental Figure S11).

Discussion

In this phase 1b trial in patients with R/R AML who were ineligible for cytotoxic chemotherapy, ven-cobi showed limited efficacy comparable to single-agent venetoclax,⁸ but with additional toxicity. The safety profile of ven-cobi was as anticipated, with no new safety signals identified for either venetoclax or cobimetinib. The observed dose-limiting toxicities (diarrhea and ejection fraction decreased) have been previously identified as risks with venetoclax and cobimetinib treatment: diarrhea has been associated with both venetoclax and cobimetinib, and ejection fraction decreased has been associated with cobimetinib. Overall, the observed PK profiles of venetoclax and cobimetinib were similar to previous findings.^{10,28}

Diarrhea was the most common AE and was observed at higher rates than previously observed with either cobimetinib²⁹ or venetoclax monotherapy⁸ (80.0% vs 61% vs 56% for allgrade diarrhea, and 36.7% vs 7% vs 6% for grade \geq 3 diarrhea, respectively). However, there are caveats which make cross-trial comparisons difficult, such as different diseases (eg, advanced solid tumors in the cobimetinib monotherapy study), and different dosing levels/schedules. The observed difference in rates of diarrhea should therefore be viewed cautiously. Although limited by small patient numbers, diarrhea appeared to be cobimetinib dose-related, with grade ≥ 3 diarrhea observed at higher rates in Cohort 4, which evaluated cobimetinib 60 mg (as opposed to cobimetinib 40 mg in Cohorts 1-3) and exceeded the cobimetinib MTD and the RP2D. While diarrhea was frequent, dose modifications/interruptions due to diarrhea were infrequent, particularly at the RP2D (all dose cohorts: 26.7%; RP2D: 14.3%); similarly, only 1 patient (3.3%; patient was in Cohort 2 [RP2D]) withdrew from treatment due to diarrhea. Though the majority of events were manageable with supportive measures, mandatory antidiarrheal prophylaxis is encouraged for future studies to improve tolerability. The frequency of other AEs, including hematologic toxicities and infections, was consistent with the known myelosuppressive effects of venetoclax and cobimetinib, and within the previously reported ranges for a similar R/R AML population.^{2,4-6,8,30,31} Clinical TLS was infrequent, as reported in other venetoclax AML trials,^{8,28} but was noted in 1 patient, where it was managed and did not lead to treatment discontinuation; it is therefore recommended that TLS prophylaxis and monitoring should continue when using venetoclax-based combinations in AML. At the proposed RP2D (venetoclax 600 mg + cobimetinib 40 mg), the safety profile was deemed manageable with the addition of risk mitigation measures for gastrointestinal toxicity.

The clinical activity of ven-cobi was limited across all dose cohorts evaluated (all dose cohorts CRc: 15.6%; RP2D CRc: 12.5%). Ven-cobi activity was comparable to that observed with single-agent venetoclax (CR/CRi 19%)8 or MEK inhibitors (trametinib: CR/CRi/MLFS/PR 20%; selumetinib: CR/CRi/PR/minor responses [MR]/unconfirmed MR 17%; binimetinib: CR/CRi 8%).¹²⁻¹⁴ Despite encouraging preclinical rationale,¹⁵ the response rates suggest no added benefit with this BCL-2/MEK inhibitor combination therapy in the clinical setting. Similarly, a previous study evaluating the combination of venetoclax-trametinibazacitidine in R/R AML demonstrated little benefit over single agent trametinib.32 However, these cross-trial comparisons are limited by differences in patient populations, such as the inclusion of patients with RAS mutations only, high-risk myelodysplastic syndromes, or unfit newly-diagnosed AML, and differences in response definitions.¹²⁻¹⁴ Further, in the current study, the design limited the ability to draw conclusions regarding achievement of transfusion independence for red blood cells or platelets; capturing this information is therefore recommended to complement efficacy data in future R/R AML studies. Overall, the observed response rates and OS reported with ven-cobi were within the expected range for an elderly or unfit R/R AML population.^{5,6,33}

Marina Y. Konopleva et al

While an RP2D was determined, due to the overall risk-benefit profile, expansion was not pursued. Of note, a short median duration of therapy (2 cycles [range 1-8]) was observed; this reflects the risk-benefit profile, including the limited quality of the responses and the added toxicity of the combination. It is not clear whether dosing optimization, such as continuous MEK inhibition or intermittent BCL-2 dosing as investigated in the ven-idasa arm, and mandatory gastrointestinal prophylaxis could improve the benefitrisk profile. Additionally, this study was conducted in an unfit R/R AML population, characterized by frailty and poor outcomes; as such, this may have had a negative impact on the risk-benefit profile.

Further, 1 potential explanation for the limited efficacy observed is insufficient target or pathway inhibition. For instance, a lack of exposure dependency or response effect in pERK and pS6 was noted. Baseline levels of pERK, which is proximal to MEK and therefore likely best reflects MEK activity,34 was associated with antileukemic responses to ven-cobi. However, pERK was only partially reduced by cobimetinib in most patients, and transiently suppressed in some patients, suggesting that MAPK inhibition was not deep nor durable. In contrast, pS6, which is known to act downstream of multiple signaling pathways, so may be a less reliable assessment of MEK/MAPK activity,35 was inhibited at later timepoints. Furthermore, while MEK inhibition has been reported to downregulate MCL-1, this was not clearly observed;¹⁸ it is unclear whether this was related to feedback loops that reactivate MAPK^{36,37} or inadequate dosing, and is difficult to conclude given the limited number of responders. Previously described resistance mechanisms such as activation of parallel signaling pathways, including PI3K/AKT, JAK/STAT, or NFKB, or upregulation of receptor tyrosine kinases may have contributed to the limited clinical activity.³⁸⁻⁴⁰ Alternatively, sample timepoints and/or collection may have affected the ability to detect PD effects.

Similar to ven-idasa,²¹ patients with low (\leq 1.5) baseline BCL-2:BCL-xL and BCL-2:MCL-1 ratios had a lower response rate to ven-cobi compared with patients with high (>1.5) baseline ratios. MCL-1 expression was most associated with nonresponse to ven-cobi, indicating that MCL-1, a known venetoclax resistance factor,²⁶ was not sufficiently inhibited by cobimetinib. Targeting MCL-1 in AML remains an unmet need, and thus alternative strategies are being explored.^{41,42}

Although an all-comer population was enrolled, rather than a *RAS* mutant population, this did not appear to drive the activity profile. Antileukemic activity was noted both in patients with (14.3%) and without (11.1%) *RAS* mutations (1/7 and 3/21 patients with baseline mutation data available, respectively). Similar results were observed with a broader definition of "signaling mutations" known to activate RAS/MAPK signaling, suggesting that RAS signaling as defined by mutational status did not define ven-cobi efficacy. However, it is important to note that as sequencing data were limited (available in 4/6 patients that achieved a response), it is difficult to delineate mutations associated with response. Importantly, pERK protein levels were elevated in patients achieving an MLFS or better response, suggesting that pretreatment MAPK activation may confer responsiveness to therapy.

Emergent TP53 mutations were observed in 16.7% of patients treated with ven-cobi, which is fewer than the 33.3% noted with

ven-idasa.²¹ This suggests that the effect of ven-idasa on TP53 mutation emergence may only be partially observed with ven-cobi, and may be a common occurrence in R/R AML, even with nontargeted therapy.43 However, the small number of responders and the short ven-cobi treatment duration may have contributed to the observed difference in the frequency of emergent TP53 mutations. In addition, emergent signaling mutations were observed in 3 patients, and clearance in 2 patients. Interestingly, clearance was observed in the context of 3 other stable signaling mutations in 1 patient, and both clearance of NRAS and emergence of PTPN11 were noted at the same timepoint in the other patient. These data indicate selective pressure to maintain MAPK signaling, through various mutations upstream of MEK. This is further supported by the observed increase in the "hallmark KRAS signaling Up" gene signature observed with ven-cobi therapy. Together these data indicate that there is strong pressure to drive RAS signaling networks in AML, particularly in the context of MEK inhibition, and suggest that effective strategies may need to target further downstream, such as MCL-1 directly, rather than indirectly through complex signaling networks. Furthermore, it was observed that pERK levels were associated with response. This, along with the increase in "hallmark KRAS signaling Up" gene signature and the emergence of signaling mutations on therapy, suggests that there may be an optimal level of pathway activation needed to achieve a response, with too little or too much RAS signaling impairing the efficacy of ven-cobi.

Although limited, analyses of VAF dynamics showed an overall decrease in VAF in responders, slight increases in patients with SD, and more extensive increases in patients with resistant disease, in line with molecular changes tracking clinical response. Inconsistent effects on RAS pathway mutations were similarly observed in R/R AML patients responding to triplet trametinib-ven-azacitidine therapy.³² Our findings suggest that AML clones harboring a signaling mutation are not uniformly sensitive to cobimetinib, and may be influenced by other factors such as the specific mutated gene, co-occurring mutations, number of signaling mutations, nongenetic factors, or extracellular signals. Notably, coculture of AML blasts with mesenchymal stromal cells consistently induced MAPK signaling in AML cells - a mechanism that contributes to mutationagnostic resistance to multiple treatment modalities.⁴⁴ However, low numbers of antileukemic responders and incomplete serial mutation data in the current study limited the ability to draw conclusions on the effect of cobimetinib on mutation dynamics, and the mechanisms of response/resistance to ven-cobi. Therefore, future studies of RAS/MAPK inhibitors in AML should carefully evaluate clonal dynamics at serial timepoints to better inform sensitivity and resistance patterns using mutations, and nongenetic approaches to assess pathway activation.

As this study was designed before venetoclax was approved for newly diagnosed unfit AML patients, all patients were venetoclaxnaïve. Thus, a limitation of this study is the inability to make conclusions regarding venetoclax retreatment. While it has been suggested that venetoclax retreatment may still be effective due to unique synergies that may exist with different venetoclax combinations,⁴⁵ future studies of BCL-2 inhibitor combinations are needed to address this data gap. Additionally, a challenge of the current study is the ability to dissect specific correlates for cobime-

tinib versus venetoclax, as some patients may have been responding to venetoclax only; a key example of this is the responses observed in patients either lacking signaling mutations or harboring *IDH* mutations. Furthermore, the correlative findings observed in the current study, such as responses in patients without signaling mutations, may not translate to patients with R/R AML previously treated with venetoclax.

Targeting RAS-activated AML clones is an area of high unmet need. Not only is the RAS/RAF/MEK/ERK signaling pathway frequently dysregulated in AML, but mutations in RAS signaling genes have been linked to resistance to newer targeted therapies including FLT3 inhibitors, IDH inhibitors, and venetoclax.^{20,46-52} Outgrowth of RAS-mutated clones has been observed at a high frequency in the relapsed setting after these agents.⁴⁶⁻⁵² As patients are likely to be treated with one of these agents earlier in their disease course, it is anticipated that there will be an increased prevalence of RAS signaling mutations in the relapsed setting; approaches that target RAS mutant clones in the R/R setting, or suppress the outgrowth of clones with RAS-activated signaling via combination therapy in the frontline setting, are therefore needed. Further, in preclinical studies, MEK inhibition led to cytostatic effects and less pronounced cytotoxic effects in comparison with downstream inhibition of the RAS-MAPK pathway.53 Other approaches that may be effective include: alternative RAS-MAPK pathway targets, such as specific RAS, ERK, or RAF mutation inhibitors⁵³⁻⁵⁵; inhibition of alternative signaling pathways⁵⁶; co-inhibition of multiple signaling molecules⁵⁷; or MEK inhibition in conjunction with sensitizing agents.58

In conclusion, despite encouraging preclinical rationale, ven-cobi showed limited preliminary efficacy, similar to single-agent venetoclax, but with added toxicity in patients with R/R AML ineligible for cytotoxic chemotherapy. The results of this study therefore suggest no added benefit with this BCL-2/MEK inhibitor combination therapy in the clinical setting. Biomarker correlatives suggest that high MCL-1 levels, *TP53* mutations, and multiple signaling mutations may contribute to resistance to ven-cobi. Future trials combining venetoclax with other MAPK pathway inhibitors are anticipated to further expand on these findings. The safety, dosing, molecular, and PD learnings from this study may therefore optimize the design and patient selection for future trials of BCL-2 and MAPK pathway inhibitor combinations.

Clinical Practice Points

- Current treatment options for patients with R/R AML are limited and are associated with poor survival outcomes.
- The combination of B-cell lymphoma-2 and MAPK/MEK inhibitors, such as venetoclax and cobimetinib, respectively, has demonstrated a promising preclinical rationale in *in vitro* AML models and *in vivo* xenograft models.
- This phase 1b trial of the venetoclax and cobimetinib combination demonstrated limited preliminary efficacy, similar to single-agent venetoclax, but with added toxicity in patients with R/R AML.
- However, the safety, dosing, molecular and pharmacodynamic learnings from this study may optimize patient selection and the

design of anticipated future trials of venetoclax in combination with MAPK pathway inhibitors.

Disclosure

MYK is a consultant for AbbVie, Inc., Genentech, Inc., and F. Hoffmann La-Roche; is an advisory board member for F. Hoffmann La-Roche; holds shares from Reata Pharmaceuticals; has received honoraria from Amgen, AbbVie, Inc., and Genentech, Inc.; and has received research funding from AbbVie, Inc., Genentech, Inc., Eli Lilly, Cellectis, Calithera, Stemline, Threshold, Flexus Biosciences, Novartis, Ablynx, and Agios. MD, MO, JW, WH, DD, and HH are employees of Genentech, Inc. and may hold Roche stock or stock options. NGD has received research funding from Daiichi Sankyo, Bristol-Myers Squibb, Pfizer, Gilead, Sevier, Genentech, Inc., Astellas, Daiichi Sankyo, AbbVie, Inc., Hanmi, Trovagene, Fate Therapeutics, Amgen, Novimmune, GlycoMimetics, Trillium, Kite Pharma, Aptose, Shattuck Labs, KAHR, ArcellX, Sanofi, Sumitomo, and ImmunoGen; and has served in a consulting or advisory role for Daiichi Sankyo, Bristol-Myers Squibb, Arog Pharmaceuticals, Pfizer, Novartis, Jazz Pharmaceuticals, Celgene, AbbVie, Inc., Astellas, Genentech, Inc., Immunogen, Servier, Syndax, Trillium, Gilead, Amgen, Shattuck Labs, ArcellX, Kite Pharma, Sumitomo, Caribou Biosciences, Sanofi, Rigel, Aptose, KURA, GlycoMimetics, and Agios. JSG has received research funding (for trials) from AbbVie, Inc., Genentech, Inc., Pfizer, Prelude, and AstraZeneca; and has served on advisory boards for AbbVie, Inc., Astellas, Bristol-Myers Squibb and Servier. BAJ is a consultant/advisor for AbbVie, Inc., Bristol-Myers Squibb, Daiichi Sankyo, Genentech, Inc., Gilead, GlycoMimetics, Jazz Pharmaceuticals, Kymera, Pfizer, Rigel, Servier, and Takeda; protocol steering committee for GlycoMimetics; data monitoring committee for Gilead; travel reimbursement/support from Rigel; and research funding to his institution from AbbVie, Inc., Amgen, Arog Pharmaceuticals, Aptose, BMS, Celgene, Daiichi Sankyo, F. Hoffmann-La Roche, Forma Therapeutics, Forty-Seven, Genentech, Inc./Roche, Gilead, GlycoMimetics, Hanmi, Immune-Onc, Incyte, Jazz Pharmaceuticals, Loxo Oncology, Pfizer, Pharmacyclics, Sigma Tau, and Treadwell. KWLY is a consultant for Astex, Bristol-Myers Squibb/Celgene, F. Hoffmann-La Roche, Novartis, Otsuka, Paladin, Pfizer, Shattuck Labs, Taiho, and Takeda; has received honorarium from AbbVie, Inc. and Novartis; and has received research funding from Astex, Forma Therapeutics, F. Hoffmann-La Roche, Genentech, Inc., Geron, Janssen, Jazz Pharmaceuticals, MedImmune, Novartis, Onconova, and Tolero. KRK, NV, SP, AT, AP, PF, and GV declare no competing financial interests. SA reports non-financial support from Roche/Genentech, Inc. during the conduct of the study; and has received personal fees from Roche Canada, Pfizer, Bristol-Myers Squibb, Palladin, and Lundbeck outside of the submitted work. GJR consulted for AbbVie, Inc., Amgen, Argenx, AstraZeneca, Bluebird Bio, Blueprint Medicines, Bristol-Myers Squibb, Caribou Biosciences, Celgene, Daiichi Sankyo, Ellipses Pharma, GlaxoSmithKline, Janssen, Jasper Pharmaceuticals, Jazz Pharmaceuticals, Molecular Partners, Novartis, Pfizer, Roche, Syndax, Takeda (IRC Chair), and Telix Pharma; and has received research funding from Janssen. DAP has received research funding from AbbVie, Inc. and has served as a consultant or advisory board member for AbbVie, Inc. and Genentech, Inc. JMB consulted for AbbVie, Inc., Bristol-Myers Squibb, Astex, Pfizer, Astellas, Taiho, and Jazz Pharmaceuticals. BLP has received research funding from Ambit Biosciences, Genentech, Inc., F. Hoffmann-La Roche, Jazz Pharmaceuticals, Novartis, Pfizer, and Cornerstone Pharmaceuticals; and is a consultant for Cornerstone Pharmaceuticals. RLO has received research support for trials from Astellas, Pfizer, Genentech, Inc., Daiichi Sankyo, and Cellectis; and consulted for AbbVie, Inc., Astellas, and Actinium. GM is a consultant for AbbVie, Inc., Celgene, Roche, Janssen, Astellas, Pfizer, and Incyte. W-JH is a former employee of Genentech, Inc. and may hold Roche stock; and is a current employee of Prelude Therapeutics. MGO is an employee of F. Hoffmann-La Roche and may hold stock or stock options. MA consulted for Amgen, AstraZeneca, Daiichi Sankyo, Syndax, GlycoMimetics, Oncoceutics, and Aptose; has received research funding from F. Hoffmann-La Roche, AstraZeneca, Amgen, Daiichi Sankyo, Jazz Pharmaceuticals, GlycoMimetics; and holds stock from Reata, Oncoceutics/Chimerix and Aptose.

CRediT authorship contribution statement

Marina Y. Konopleva: Conceptualization, Methodology, Investigation, Formal analysis, Writing - original draft, Writing - review & editing. Monique Dail: Conceptualization, Methodology, Investigation, Formal analysis, Writing - original draft, Writing review & editing. Naval G. Daver: Conceptualization, Methodology, Investigation, Formal analysis, Writing - original draft, Writing - review & editing. Jacqueline S. Garcia: Investigation, Formal analysis, Writing - review & editing. Brian A. Jonas: Investigation, Formal analysis, Writing - review & editing. Karen W.L. Yee: Investigation, Formal analysis, Writing - review & editing. Kevin R. Kelly: Investigation, Formal analysis, Writing - review & editing. Norbert Vey: Investigation, Formal analysis, Writing - review & editing. Sarit Assouline: Investigation, Formal analysis, Writing review & editing. Gail J. Roboz: Investigation, Formal analysis, Writing - review & editing. Stefania Paolini: Investigation, Formal analysis, Writing - review & editing. Daniel A. Pollyea: Investigation, Formal analysis, Writing - review & editing. Agostino Tafuri: Investigation, Formal analysis, Writing - review & editing. Joseph M. Brandwein: Investigation, Formal analysis, Writing review & editing. Arnaud Pigneux: Investigation, Formal analysis, Writing - review & editing. Bayard L. Powell: Investigation, Formal analysis, Writing - review & editing. Pierre Fenaux: Investigation, Formal analysis, Writing - review & editing. Rebecca L. Olin: Investigation, Formal analysis, Writing - review & editing. Giuseppe Visani: Investigation, Formal analysis, Writing - review & editing. Giovanni Martinelli: Investigation, Writing - review & editing. Maika Onishi: Conceptualization, Methodology, Investigation, Formal analysis, Writing - original draft, Writing - review & editing. Jue Wang: Conceptualization, Methodology, Formal analysis, Writing - original draft, Writing - review & editing. Weize Huang: Formal analysis, Writing - original draft, Writing - review & editing. Diana R. Dunshee: Formal analysis, Writing - original draft, Writing - review & editing. Habib Hamidi: Formal analysis, Writing - original draft, Writing - review & editing. Marion G.

Marina Y. Konopleva et al

Ott: Conceptualization, Methodology, Formal analysis, Writing – review & editing. **Wan-Jen Hong:** Conceptualization, Methodology, Investigation, Formal analysis, Writing – review & editing. **Michael Andreeff:** Conceptualization, Methodology, Investigation, Formal analysis, Writing – original draft, Writing – review & editing.

Acknowledgments

Venetoclax is being developed in a collaboration between Genentech, Inc. and AbbVie, Inc. Genentech, Inc. provided financial support for the study and participated in the design, study conduct, analysis, and interpretation of data, as well as the writing, review, and approval of the manuscript. We thank the participating patients and their families, investigators, and trial staff. Denise Cheung, B. Pharm, of Genentech, Inc., provided support in analysis of the data, and Cherie Green and Connie Ma provided flow cytometry and data analysis support. Third-party medical writing assistance, under the direction of Maika Onishi, was provided by Roisin Weaver, MSc, of Ashfield MedComms, an Inizio company, and was funded bv F. Hoffmann-La Roche Ltd.

Supplementary materials

Supplementary material associated with this article can be found, in the online version, at doi:10.1016/j.clml.2024.01.007.

References

- National Cancer Institute. Cancer stat facts: leukemia acute myeloid leukemia (AML). 2020. Accessed: March 17, 2023. Available at: https://seer.cancer.gov/ statfacts/html/amyl.html.
- Appelbaum FR, Gundacker H, Head DR, et al. Age and acute myeloid leukemia. Blood. 2006;107:3481–3485. doi:10.1182/blood-2005-09-3724.
- Ganzel C, Sun Z, Cripe LD, et al. Very poor long-term survival in past and more recent studies for relapsed AML patients: the ECOG-ACRIN experience. *Am J Hematol.* 2018;93:1074–1081. doi:10.1002/ajh.25162.
- Ravandi F, Ritchie EK, Sayar H, et al. Vosaroxin plus cytarabine versus placebo plus cytarabine in patients with first relapsed or refractory acute myeloid leukaemia (VALOR): a randomised, controlled, double-blind, multinational, phase 3 study. *Lancet Oncol.* 2015;16:1025–1036. doi:10.1016/s1470-2045(15)00201-6.
- Roboz GJ, Rosenblat T, Arellano M, et al. International randomized phase III study of elacytarabine versus investigator choice in patients with relapsed/refractory acute myeloid leukemia. *J Clin Oncol.* 2014;32:1919–1926. doi:10.1200/jco.2013. 52.8562.
- Stahl M, DeVeaux M, Montesinos P, et al. Hypomethylating agents in relapsed and refractory AML: outcomes and their predictors in a large international patient cohort. *Blood Adv.* 2018;2:923–932. doi:10.1182/bloodadvances.2018016121.
- Souers AJ, Leverson JD, Boghaert ER, et al. ABT-199, a potent and selective BCL-2 inhibitor, achieves antitumor activity while sparing platelets. *Nat Med.* 2013;19:202–208. doi:10.1038/nm.3048.
- Konopleva M, Pollyea DA, Potluri J, et al. Efficacy and biological correlates of response in a phase II study of venetoclax monotherapy in patients with acute myelogenous leukemia. *Cancer Discov.* 2016;6:1106–1117. doi:10.1158/ 2159-8290.Cd-16-0313.
- Hoeflich KP, Merchant M, Orr C, et al. Intermittent administration of MEK inhibitor GDC-0973 plus PI3K inhibitor GDC-0941 triggers robust apoptosis and tumor growth inhibition. *Cancer Res.* 2012;72:210–219. doi:10.1158/0008-5472. Can-11-1515.
- Genentech. U.S. prescribing information: COTELLIC (cobimetinib). 2022. Accessed: March 17, 2023. Available at: https://www.gene.com/download/pdf/ cotellic_prescribing.pdf.
- European Medicines Agency. Summary of product characteristics: COTELLIC (cobimetinib). 2023. Accessed: March 17, 2023. Available at: https://www.ema.europa.eu/en/documents/product-information/ cotellic-epar-product-information_en.pdf.
- Borthakur G, Popplewell L, Boyiadzis M, et al. Activity of the oral mitogenactivated protein kinase kinase inhibitor trametinib in RAS-mutant relapsed or refractory myeloid malignancies. *Cancer*. 2016;122:1871–1879. doi:10.1002/cncr. 29986.
- 13. Jain N, Curran E, Iyengar NM, et al. Phase II study of the oral MEK inhibitor selumetinib in advanced acute myelogenous leukemia: a University of Chicago

phase II consortium trial. Clin Cancer Res. 2014;20:490-498. doi:10.1158/1078-0432.Ccr-13-1311.

- Maiti A, Naqvi K, Kadia TM, et al. Phase II trial of MEK inhibitor binimetinib (MEK162) in RAS-mutant acute myeloid leukemia. *Clin Lymphoma Myeloma Leuk*. 2019;19:142–148.e141. doi:10.1016/j.clml.2018.12.009.
- Han L, Zhang Q, Dail M, et al. Concomitant targeting of BCL2 with venetoclax and MAPK signaling with cobimetinib in acute myeloid leukemia models. *Haema-tologica*. 2020;105:697–707. doi:10.3324/haematol.2018.205534.
- Milella M, Precupanu CM, Gregorj C, et al. Beyond single pathway inhibition: MEK inhibitors as a platform for the development of pharmacological combinations with synergistic anti-leukemic effects. *Curr Pharm Des*. 2005;11:2779–2795. doi:10.2174/1381612054546842.
- Milella M, Estrov Z, Kornblau SM, et al. Synergistic induction of apoptosis by simultaneous disruption of the BCL-2 and MEK/MAPK pathways in acute myelogenous leukemia. *Blood*. 2002;99:3461–3464. doi:10.1182/blood.v99. 9.3461.
- Konopleva M, Milella M, Ruvolo P, et al. MEK inhibition enhances ABT-737induced leukemia cell apoptosis via prevention of ERK-activated MCL-1 induction and modulation of MCL-1/BIM complex. *Leukemia*. 2012;26:778–787. doi:10. 1038/leu.2011.287.
- Kaufmann SH, Karp JE, Svingen PA, et al. Elevated expression of the apoptotic regulator MCL-1 at the time of leukemic relapse. *Blood.* 1998;91:991–1000. doi:10.1182/blood.V91.3.991.
- Zhang Q, Riley-Gillis B, Han L, et al. Correction: activation of RAS/MAPK pathway confers MCL-1 mediated acquired resistance to BCL-2 inhibitor venetoclax in acute myeloid leukemia. *Signal Transduct Target Ther.* 2022;7:110. doi:10. 1038/s41392-022-00958-4.
- 21. Daver NG, Dail M, Garcia JS, et al. Venetoclax and idasanutlin in relapsed/refractory AML: a nonrandomized, open-label phase 1b trial. *Blood*. 2023;141:1265–1276. doi:10.1182/blood.2022016362.
- Ascierto PA, McArthur GA, Dréno B, et al. Cobimetinib combined with vemurafenib in advanced BRAF(V600)-mutant melanoma (coBRIM): updated efficacy results from a randomised, double-blind, phase 3 trial. *Lancet Oncol.* 2016;17:1248–1260. doi:10.1016/s1470-2045(16)30122-x.
- Arber DA, Orazi A, Hasserjian R, et al. The 2016 revision to the World Health Organization classification of myeloid neoplasms and acute leukemia. *Blood.* 2016;127:2391–2405. doi:10.1182/blood-2016-03-643544.
- Cheson BD, Bennett JM, Kopecky KJ, et al. Revised recommendations of the international working group for diagnosis, standardization of response criteria, treatment outcomes, and reporting standards for therapeutic trials in acute myeloid leukemia. J Clin Oncol. 2003;21:4642–4649. doi:10.1200/jco.2003.04.036.
- He J, Abdel-Wahab O, Nahas MK, et al. Integrated genomic DNA/RNA profiling of hematologic malignancies in the clinical setting. *Blood*. 2016;127:3004–3014. doi:10.1182/blood-2015-08-664649.
- Liu J, Chen Y, Yu L, Yang L. Mechanisms of venetoclax resistance and solutions. Front Oncol. 2022;12:1005659. doi:10.3389/fonc.2022.1005659.
- Liberzon A, Birger C, Thorvaldsdóttir H, Ghandi M, Mesirov JP, Tamayo P. The molecular signatures database (MSigDB) hallmark gene set collection. *Cell Syst.* 2015;1:417–425. doi:10.1016/j.cels.2015.12.004.
- Genentech. U.S. prescribing information: VENCLEXTA® (venetoclax). 2022. Accessed: June 12, 2023. Available at: https://www.rxabbvie.com/pdf/venclexta. pdf.
- Rosen LS, LoRusso P, Ma WW, et al. A first-in-human phase I study to evaluate the MEK1/2 inhibitor, cobimetinib, administered daily in patients with advanced solid tumors. *Invest New Drugs*, 2016;34:604–613. doi:10.1007/s10637-016-0374-3.
- Aldoss I, Yang D, Aribi A, et al. Efficacy of the combination of venetoclax and hypomethylating agents in relapsed/refractory acute myeloid leukemia. *Haematologica*. 2018;103:e404–e407. doi:10.3324/haematol.2018.188094.
- Tenold ME, Moskoff BN, Benjamin DJ, et al. Outcomes of adults with relapsed/refractory acute myeloid leukemia treated with venetoclax plus hypomethylating agents at a comprehensive cancer center. *Front Oncol.* 2021;11:649209. doi:10.3389/fonc.2021.649209.
- Desikan SP, Ravandi F, Pemmaraju N, et al. A phase II study of azacitidine, venetoclax, and trametinib in relapsed or refractory acute myeloid leukemia harboring RAS pathway-activating mutations. *Acta Haematol.* 2022;145:529–536. doi:10. 1159/000525566.
- Itzykson R, Thépot S, Berthon C, et al. Azacitidine for the treatment of relapsed and refractory AML in older patients. *Leuk Res.* 2015;39:124–130. doi:10.1016/j. leukres.2014.11.009.
- Zhang W, Liu HT. MAPK signal pathways in the regulation of cell proliferation in mammalian cells. *Cell Res.* 2002;12:9–18. doi:10.1038/sj.cr.7290105.
- Fenton TR, Gout IT. Functions and regulation of the 70kDa ribosomal S6 kinases. Int J Biochem Cell Biol. 2011;43:47–59. doi:10.1016/j.biocel.2010.09.018.

- Amit I, Citri A, Shay T, et al. A module of negative feedback regulators defines growth factor signaling. *Nat Genet.* 2007;39:503–512. doi:10.1038/ng1987.
- Shin SY, Rath O, Choo SM, et al. Positive- and negative-feedback regulations coordinate the dynamic behavior of the RAS-RAF-MEK-ERK signal transduction pathway. J Cell Sci. 2009;122:425–435. doi:10.1242/jcs.036319.
- Gille H, Downward J. Multiple RAS effector pathways contribute to G(1) cell cycle progression. J Biol Chem. 1999;274:22033–22040. doi:10.1074/jbc.274.31. 22033.
- Kun E, Tsang YTM, Ng CW, Gershenson DM, Wong KK. MEK inhibitor resistance mechanisms and recent developments in combination trials. *Cancer Treat Rev.* 2021;92:102137. doi:10.1016/j.ctrv.2020.102137.
- Shi H, Hugo W, Kong X, et al. Acquired resistance and clonal evolution in melanoma during BRAF inhibitor therapy. *Cancer Discov.* 2014;4:80–93. doi:10. 1158/2159-8290.Cd-13-0642.
- Carter BZ, Mak PY, Tao W, et al. Combined inhibition of BCL-2 and MCL-1 overcomes BAX deficiency-mediated resistance of TP53-mutant acute myeloid leukemia to individual BH3 mimetics. *Blood Cancer J.* 2023;13:57. doi:10.1038/s41408-023-00830-w.
- Carter BZ, Mak PY, Tao W, et al. Targeting MCL-1 dysregulates cell metabolism and leukemia-stroma interactions and re-sensitizes acute myeloid leukemia to BCL-2 inhibition. *Haematologica*. 2020;107:58–76. doi:10.3324/haematol.2020. 260331.
- Alwash Y, Khoury JD, Tashakori M, et al. Development of TP53 mutations over the course of therapy for acute myeloid leukemia. *Am J Hematol.* 2021;96:1420– 1428. doi:10.1002/ajh.26314.
- Konopleva M, Konoplev S, Hu W, Zaritskey AY, Afanasiev BV, Andreeff M. Stromal cells prevent apoptosis of AML cells by up-regulation of anti-apoptotic proteins. *Leukemia*. 2002;16:1713–1724. doi:10.1038/sj.leu.2402608.
- Daver N, Perl AE, Maly J, et al. Venetoclax plus gilteritinib for FLT3-mutated relapsed/refractory acute myeloid leukemia. J Clin Oncol. 2022;40:4048–4059. doi:10.1200/jco.22.00602.
- McMahon CM, Ferng T, Canaani J, et al. Clonal selection with RAS pathway activation mediates secondary clinical resistance to selective FLT3 inhibition in acute myeloid leukemia. *Cancer Discov.* 2019;9:1050–1063. doi:10.1158/ 2159-8290.Cd-18-1453.
- Smith CC, Levis MJ, Perl AE, Hill JE, Rosales M, Bahceci E. Molecular profile of FLT3-mutated relapsed/refractory patients with AML in the phase 3 ADMIRAL study of gilteritinib. *Blood Adv.* 2022;6:2144–2155. doi:10.1182/bloodadvances. 2021006489.
- Wang F, Morita K, DiNardo CD, et al. Leukemia stemness and co-occurring mutations drive resistance to IDH inhibitors in acute myeloid leukemia. *Nat Commun.* 2021;12:2607. doi:10.1038/s41467-021-22874-x.
- Stein EM, DiNardo CD, Fathi AT, et al. Molecular remission and response patterns in patients with mutant-IDH2 acute myeloid leukemia treated with enasidenib. *Blood.* 2019;133:676–687. doi:10.1182/blood-2018-08-869008.
- Maiti A, Rausch CR, Cortes JE, et al. Outcomes of relapsed or refractory acute myeloid leukemia after frontline hypomethylating agent and venetoclax regimens. *Haematologica*. 2021;106:894–898. doi:10.3324/haematol.2020.252569.
- DiNardo CD, Tiong IS, Quaglieri A, et al. Molecular patterns of response and treatment failure after frontline venetoclax combinations in older patients with AML. *Blood.* 2020;135:791–803. doi:10.1182/blood.2019003988.
- Cherry EM, Abbott D, Amaya M, et al. Venetoclax and azacitidine compared with induction chemotherapy for newly diagnosed patients with acute myeloid leukemia. *Blood Adv.* 2021;5:5565–5573. doi:10.1182/bloodadvances. 2021005538.
- Tambe M, Karjalainen E, Vähä-Koskela M, et al. Pan-RAF inhibition induces apoptosis in acute myeloid leukemia cells and synergizes with BCL2 inhibition. *Leukemia*. 2020;34:3186–3196. doi:10.1038/s41375-020-0972-0.
- Popescu B, Stahlhut C, Tarver TC, et al. Allosteric SHP2 inhibition increases apoptotic dependency on BCL2 and synergizes with venetoclax in *FLT3-* and *KIT*mutant AML. *Cell Rep Med.* 2023;4:101290. doi:10.1016/j.xcrm.2023.101290.
- Khoury JD, Tashakori M, Yang H, et al. Pan-RAF inhibition shows anti-leukemic activity in *RAS*-mutant acute myeloid leukemia cells and potentiates the effect of sorafenib in cells with *FLT3* mutation. *Cancers*. 2020;12:3511. doi:10.3390/ cancers12123511.
- Bouligny IM, Maher KR, Grant S. Augmenting venetoclax activity through signal transduction in AML. J Cell Signal. 2023;4:1–12. doi:10.33696/signaling.4.085.
- Yen I, Shanahan F, Lee J, et al. ARAF mutations confer resistance to the RAF inhibitor belvarafenib in melanoma. *Nature*. 2021;594:418–423. doi:10.1038/ s41586-021-03515-1.
- Decroocq J, Birsen R, Montersino C, et al. RAS activation induces synthetic lethality of MEK inhibition with mitochondrial oxidative metabolism in acute myeloid leukemia. *Leukemia*. 2022;36:12370 11252. doi:10.1038/s41375-022-01541-0.